379

Anticoagulation for non-valvular atrial fibrillation: new anticoagulant agents

Non-valvüler atriyal fibrilasyonda antikoagülasyon: Yeni antikoagülan ilaçlar

Alper Kepez, Okan Erdoğan

Department of Cardiology, Faculty of Medicine, Marmara University İstanbul-Turkey

Abstract

Atrial fibrillation (AF) is a common cardiac arrhythmia and it is associated with systemic thromboembolism. Until recently, vitamin K antagonists (VKA) such as warfarin were the only available oral anticoagulant therapy for prevention of stroke and systemic embolism in AF. Limitations of VKA therapy have prompted researchers to search for novel anticoagulant drugs, which do not necessitate coagulation monitoring due to their more predictable pharmacokinetic profile. Large-scale phase III trials have been completed for some of these drugs and 'U.S. Food and Drug Administration (FDA)' approved dabigatran and rivaroxaban for prevention of systemic embolism in non-valvular AF patients. In this review, we will first focus on pharmacodynamic and pharmacokinetic profiles of these medications and then try to overview clinical trial results. We will also try to mention the current controversies regarding the clinical application of these drugs. (*Anadolu Kardiyol Derg 2013; 13: 379-84*) **Key words**: Atrial fibrillation, stroke, systemic embolism, anticoagulation, vitamin K antagonists, novel anticoagulant drugs

ÖZET

Atriyal fibrilasyon (AF) sistemik tromboembolizm ile ilişkili olan nispeten sık bir aritmidir. Yakın zamana kadar AF ile ilişkili sistemik embolizm ve inmenin engellenmesinde yalnızca varfarin gibi K vitamini antagonistleri kullanılmıştır. Buna karşın K vitamini antagonistlerinin kısıtlılıkları araştırmacıları farmakokinetik profilleri daha stabil olan ve koagülasyon monitorizasyonu gerektirmeyen yeni antikoagülan ilaçların geliştirilmesine teşvik etmiştir. Bu ilaçların bazılarının faz 3 klinik çalışmaları sonuçlanmış olup 'Amerikan İlaç ve Gıda Dairesi (FDA)' tarafından dabigatran ve rivaroksabanın non-valvüler AF hastalarında inme ve sistemik embolizm profilaksisinde kullanımı onaylanmıştır. Bu derleme yazıda öncelikle yeni antikoagülan ilaçların farmakodinamik ve farmakokinetik özellikleri üzerinde durulacak, daha sonra klinik çalışma sonuçları özetlenecektir. Ayrıca, günümüzde bu ilaçların klinik pratikte kullanımındaki çekinceli durumlara da değinilmeye çalışılacaktır. (Anadolu Kardiyol Derg 2013; 13: 379-84)

Anahtar kelimeler: Atriyal fibrilasyon, inme, sistemik embolizm, antikoagülasyon, K vitamini antagonistleri, yeni antikoagülan ilaçlar

Introduction

Atrial fibrillation (AF) is a common cardiac arrhythmia and it is associated with systemic thromboembolism (1, 2). Loss of atrial mechanical function and atrial dilatation predispose to thrombus formation and certain risk factors increase the risk. Antithrombotic therapy is the established method for both primary and secondary prevention of stroke or systemic embolization in patients with AF (3). Until recently, vitamin K antagonists (VKA) such as warfarin were the only available oral anticoagulant therapy for stroke prevention in AF. In six clinical trials of patients with AF at high risk of stroke, adjusted dose warfarin therapy decreased the risk of stroke or systemic embolism by 62% compared with placebo (4). However, warfarin therapy is difficult to use in clinical practice. VKA therapy has a wide variation regarding its dose-response profile, and potentially interacts with diets and other medications (5). Inadequate

Address for Correspondence/Yazışma Adresi: Dr. Okan Erdoğan, Marmara Üniversitesi Tıp Fakültesi, Kardiyoloji Anabilim Dalı, Kadıköy, İstanbul-*Türkiye* Phone: +90 216 336 02 12 Fax: +90 216 657 09 65 E-mail: okanerdogan@yahoo.com Accepted Date/Kabul Tarihi: 01.02.2013 Available Online Date/Çevrimiçi Yayın Tarihi: 11.04.2013 © Telif Hakkı 2013 AVES Yayıncılık Ltd. Şti. - Makale metnine www.anakarder.com web sayfasından ulaşılabilir. © Copyright 2013 by AVES Yayıncılık Ltd. - Available online at www.anakarder.com doi:10.5152/akd.2013.109



monitoring will lead to either ineffective protection or excessive anticoagulation associated with increased bleeding complications. In a cohort of patients with AF receiving warfarin who were ≥ 65 years of age, the rate of intracranial hemorrhage was 2.5% (6). In addition, 17% of first admissions for intracranial hemorrhage have been reported to be associated with anticoagulant therapy and 98% of these patients have been reported to be on warfarin treatment (7). VKA therapy also has a slow onset of action and bridging with heparin or low molecular weight heparin is necessary until anticoagulation level becomes adequate as assessed with INR value (5). An ideal anticoagulant should have a predictable dose response curve and kinetics. It should be administered in fixed doses (preferentially single daily dose) without the need for routine coagulation monitoring. It should also have minimal or no interaction with food and other drugs. Limitations of VKA therapy have prompted researchers to search for ideal anticoagulant and we now have an array of new anticoagulants that act by directly inhibiting thrombin or factor Xa (8, 9). These drugs have a more predictable pharmacokinetic profile than the VKA in addition to no need for coagulation monitoring. Although VKA's are still the only available oral anticoagulant agents for stroke and systemic embolism prevention in patients with valvular AF, new anticoagulant agents have shown promise as an alternative to VKA in patients with non-valvular AF.

In this review, we aimed to focus on new oral anticoagulant agents that have been tested in phase III randomized trials for prevention of stroke and systemic embolization in patients with non-valvular AF.

Oral direct thrombin inhibitors

Dabigatran

Dabigatran etexilate is available as an oral pro drug that is converted to dabigatran after absorption. It is a potent reversible direct thrombin inhibitor (8). It binds to clot-bound and free thrombin with high affinity and specificity (8). The bioavailability of dabigatran etexilate ranges from 3% to 7% and the absorption is facilitated by the presence of an acidic milieu (10). Commercially available formulation of dabigatran etexilate contains tartaric acid which allows a steady absorption of the drug despite fluctuations in enteric pH (10). Presence of tartaric acid in the formulation is also responsible for dyspepsia, which is a common side effect of this drug (11). Time to peak concentration occurs 1.5 to 3 hours after oral administration and food intake on absorption increases the time to peak concentration (10). Approximately 80% of the administered drug is excreted in the urine and the remaining amount undergoes conjugation and glucuronidation within the liver (11). The elimination half-life is 12 to 17 hours after multiple doses in healthy patients with normal renal function and half-life is prolonged in patients with renal dysfunction (10). The metabolism of dabigatran is independent of cytochrome p450 and there is less interaction with other medications compared with warfarin (11). Unlike ximelagatran there is no reported hepatotoxicity with this agent (12). These pharmacodynamic and pharmacokinetic properties of dabigatran allow predictable anticoagulant effect with this agent and it is possible to be used with fixed doses without need for coagulation monitoring (10,11). Dabigatran etexilate, but not dabigatran, is a substrate for p-glycoprotein (P-gp) and co-administration of P-gp inducers (rifampicin or some antiepileptic drugs) or inhibitors (azole-antimycotics, immunosuppressants, human immunodeficiency virus protease inhibitors) may alter plasma concentrations of dabigatran (11, 13). Verapamil has also been reported to increase plasma concentration of dabigatran especially if immediate-release verapamil is given 1 hour before dabigatran (14). It is suggested that verapamil should be given 2 hours after dabigatran so that the interaction between these two drugs will be minimal (14). Dabigatran therapy was compared with warfarin in the phase III 'Randomized Evaluation of Long-Term Anticoagulation Therapy (RE-LY)' trial which enrolled 18.113 patients with non-valvular atrial fibrillation (15). The mean age of the patients was 71 years. 63.6% of the study population was men and half of the patients received long-term oral anticoagulation with VKA. The mean CHADS₂ score was 2.1. Patients were assigned to two fixed doses of dabigatran (110 mg or 150 mg twice daily) in a blinded fashion or to adjusted-dose warfarin in an open label fashion. The median duration of the follow-up period was 2 years. Rates of stroke or systemic embolism were 1.69% per year in the warfarin group, as compared with 1.53% per year in the group that received 110 mg bid of dabigatran (p<0.001 for noninferiority) and 1.11% per year in the group that received 150 mg bid of dabigatran (p<0.001 for superiority). Rate of major bleeding was similar between dabigatran 150 mg bid and warfarin groups (3.11% vs 3.36% respectively, p=0.31) whereas dabigatran 110 mg bid was associated with significantly lower rates of major bleeding (2.71% vs 3.36% respectively, p=0.003). There were significantly lower rates of hemorrhagic stroke in both 110 mg bid and 150 mg bid dabigatran groups compared with warfarin. The mortality rates were similar. However, there was a nonsignificant increase in myocardial infarction with dabigatran compared with warfarin (16). Dyspepsia was significantly more common with dabigatran and there was no significant difference between dabigatran and warfarin groups regarding the rate of elevation of serum aspartate aminotransferase or alanine aminotransferase levels. Subgroup analysis of data from the RE-LY was performed in patients with prior transient ischemic attack (TIA) or stroke; patients undergoing cardioversion; patients with low, moderate and high CHADS₂ score and elderly patients (17-21). Similar results were obtained from subgroup analysis of patients with previous stroke or transient ischemic attack; 150 mg bid dabigatran was superior and 110 mg dabigatran bid was non-inferior compared with warfarin for prevention of stroke or systemic embolism (18). In the subgroup analysis of patients who had undergone cardioversion during the trial, there were no significant differences between both doses of dabigatran and warfarin groups regarding stroke and major bleeding rates (19). Patients with higher CHADS₂ scores were observed to have higher rates of stroke or systemic embolism, major bleeding or vascular and total mortality in each treatment group in RE-LY trial. However, effects of 110 mg bid dabigatran and 150 mg bid dabigatran were similar with general study population and

there was no significant difference in subgroups defined by CHADS₂ scores (20). In another subgroup analysis performed to evaluate the effect of age on dabigatran therapy, both doses of dabigatran were found to be associated with lower risk of intracranial and extracranial bleeding compared with warfarin in patients aged <75 years. In patients ≥75 years of age, intracranial bleeding risk was still lower however, extracranial bleeding risk was similar or higher with both doses of dabigatran compared with warfarin (21). Dabigatran therapy was also indirectly compared with dual-antiplatelet therapy (ASA plus clopidogrel) in patients with AF who cannot use warfarin (22). Both doses of dabigatran therapy were estimated to reduce the risk of all stroke significantly compared with dual-antiplatelet therapy without increasing the rates of intracranial or extracranial hemorrhage (22). The US Food and Drug Administration (FDA) has approved 150 mg bid dose for the prevention of stroke and systemic embolism in patients with non-valvular AF (9). Accordingly, dose should be reduced to 75 mg bid for patients with renal insufficiency. European Medicine Evaluation Agency (EMEA) approved both 150 mg bid and 110 mg bid and suggested 110 mg bid dosage for elderly patients, for patients who use verapamil and for patients with high bleeding risk such as those with moderate renal impairment (creatinine clearance 30-50 mL/min) (9,13). Dabigatran is not advised to be used in patients with severe renal insufficiency (creatinine clearance <30 mL/min) (9). Dabigatran (Pradaxa) has also been licensed by Turkish Ministry of Health for the prevention of stroke and systemic embolism in patients with non-valvular AF who have an indication for anticoagulation according to their CHA₂DS₂-VASc scores. There is no specific antidote to reverse the anticoagulant effects of dabigatran. Dabigatran etexilate has been advised to be discontinued at least 24 hours before invasive procedures and at least 48 hours before procedures associated with a high risk of bleeding (14). The management of bleeding complications in patients receiving dabigatran etexilate should be individualized. In most patients with normal renal function, discontinuation of the drug will be sufficient (14). Transfusion of erythrocytes or fresh frozen plasma may be required; however, fresh-frozen plasma does not reverse the anticoagulant effect of dabigatran. If these measures fail to control bleeding, the use of hemodialysis or administration of nonspecific pro-hemostatic agents such as activated prothrombin complex concentrate may be considered (23).

Oral factor Xa inhibitors

Rivaroxaban

Rivaroxaban is an oral direct factor Xa inhibitor with a competitive and reversible binding effect to factor Xa. Rivaroxaban has an oral bioavailability of 60-80 % (17). Its half-life is reported to be between 5-9 h in young people and between 11-13 h in the elderly (13). Time to peak plasma concentration is approximately 3 hours after administration (13). Food results in delayed but increased absorption; therefore, therapeutic dosages of rivaroxaban are recommended to be taken with meals (13). Two-thirds of the drug undergoes metabolic degradation in the liver and onethird is eliminated renally as unchanged (9). Rivaroxaban is metabolized by the liver through oxidative and hydrolytic pathways catalyzed by cytochrome P450 (CYP) enzymes and is a substrate for transport P-gp (13). Therefore, rivaroxaban may interact with drugs that interact with CYP3A4 and P-gp. Rivaroxaban has been reported to bind both to free factor Xa and to factor Xa in prothrombinase complex without the need of antithrombin as a cofactor (24). There is no need for routine coagulation monitoring with rivaroxaban; however, this drug has been reported to prolong activated partial thromboplastin time (aPTT) and prothrombin time (PT) (24). Similar to dabigatran there is no specific antidote to reverse the effects of rivaroxaban (8). It is not possible to remove rivaroxaban with dialysis because it is highly bound to plasma proteins (8). In the case of overdose or bleeding, rivaroxaban therapy should be stopped and supportive care should be considered. In the case of overdose, activated charcoal may also be used in order to reduce absorption of rivaroxaban (8). "The Rivaroxaban once daily direct factor Xa inhibition compared with vitamin K antagonism for prevention of stroke and embolism trial in atrial fibrillation (ROCKET-AF)" was a phase III, double-blind and double-dummy designed study to assess the efficacy and safety of rivaroxaban compared with adjusted-dose warfarin in patients with non-valvular AF (25). A total of 14.264 patients with nonvalvular AF who were at increased risk for stroke were assigned to either rivaroxaban (at a daily dose of 20 mg) or doseadjusted warfarin. Patients with renal insufficiency (estimated creatinine clearance 30-49 mL/min) received 15 mg of rivaroxaban daily. Patients with a history of prior stroke, TIA or systemic embolism, or with two or more of the following risk factors were included into trial: clinical heart failure and/or left ventricular ejection fraction \leq 35%, hypertension, age \geq 75 years, or diabetes mellitus. Mean CHADS₂ score of patients was 3.5 and 55% of patients had a history of previous stroke, systemic embolism, or TIA. Rivaroxaban was found to be noninferior but not superior to warfarin for the primary end point of stroke or systemic embolism. Rivaroxaban therapy was associated with significantly lower rates of intracranial hemorrhage and fatal bleeding; however, there were significantly more patients with major bleeding from a gastrointestinal site in the rivaroxaban group. There were also more patients with hemoglobin fall of ≥ 2 g/dL and those who needed transfusion in the rivaroxaban group (25). Based on these observations, rivaroxaban was approved by FDA for the prevention of stroke and systemic embolism in patients with nonvalvular AF at a dose of 20 mg od (15 mg od if creatinine clearance is 15-50 mL/min) (13). It is recommended to be taken with evening meals (13). Rivaroxaban (Xarelto) has also been licensed by Turkish Ministry of Health for the prevention of stroke and systemic embolism in patients with non-valvular AF who have an indication for anticoagulation.

Apixaban

Apixaban is also an oral, direct and reversible factor Xa inhibitor with an oral bioavailability of 50% (17). Its absorption

has been reported to be independent of food administration (26). It has a half-life of 9-14 h in healthy subjects and it reaches its peak plasma concentration 3 hours after oral administration (17). Similar to rivaroxaban, much of the drug is metabolized in the liver with a cytochrome P450-dependent way and apixaban is also a substrate for transport P-gp (8). Much of the drug is removed from the body with intestinal excretion via the feces and approximately 25% of the drug is eliminated via the kidneys (27). Apixaban is highly bound to plasma proteins and it has a low distributing volume (13). Similar to rivaroxaban, apixaban binds both to free factor Xa and to factor Xa in prothrombinase complex (17). Multiple elimination pathways of apixaban may give this agent an advantage for the application in patients with renal or hepatic dysfunction (13). Two phase III studies have evaluated apixaban for the prevention of stroke and systemic embolism in AF patients. "The Apixaban Versus Acetylsalicylic Acid (ASA) to Prevent Stroke in Atrial Fibrillation Patients Who Have Failed or Are Unsuitable for Vitamin K Antagonist Treatment (AVERROES)" study compared apixaban with ASA for the prevention of stroke or systemic embolism in a population of AF patients who are not on VKA prophylaxis (28). In a doubleblinded design, 5599 patients were randomly assigned to receive apixaban (at a dose of 5 mg twice daily) or aspirin (81 to 324 mg per day) and mean follow up period was 1.1 years. The study was terminated early due to clear observed benefit in favor of apixaban with significantly lower rates of primary outcome events compared with ASA (1.6% vs 3.7% per year, respectively, p <0.001). Mortality rate was also lower for apixaban; however difference was not statistically significant (3.5% vs 4.4% per year, respectively, p=0.07). There was no significant difference between apixaban and ASA groups regarding major bleeding (1.4% vs 1.2% per year, respectively, p=0.57). The risk of a first hospitalization for cardiovascular causes was reduced with apixaban as compared with aspirin (12.6% vs. 15.9% per year, p<0.001) (28). "Apixaban for Reduction in Stroke and Other Thromboembolic Events in Atrial Fibrillation (ARISTOTLE)" study compared apixaban (at a dose of 5 mg bid) with dose adjusted warfarin in 18.201 patients with AF and at least one additional risk factor for stroke (29). The ARISTOTLE was a double-blind, non-inferiority trial and the primary outcome was ischemic or hemorrhagic stroke or systemic embolism. Secondary objectives were: testing for superiority with respect to the primary outcome and to define the rates of major bleeding and death from any cause. The mean CHADS₂ score for patients in the ARISTOTLE trial was 2.1, with less than 20% of patients having a prior stroke, transient ischemic attack or systemic embolism. The median duration of follow-up was 1.8 years. Apixaban therapy was found to be superior to warfarin for prevention of primary outcome (1.27% vs 1.60% per year, respectively, p <0.001 for noninferiority; p=0.01 for superiority). Apixaban therapy was associated with significantly lower rates of major bleeding and death from any cause (2.13% vs 3.09% per year, p <0.001 and 3.52% vs 3.94% per year, p=0.047). There were also fewer myocardial infarction and gastrointestinal bleeding events in patients assigned to apixaban therapy but the differences were not statistically significant (p=0.37 for each). Apixaban has not yet been approved by FDA for prevention of stroke or systemic embolization in AF.

Controversial issues regarding clinical application of novel anticoagulant agents

New oral anticoagulant agents seem to be relatively safe and simplify anticoagulation for the prevention of stroke and systemic embolism in patients with non-valvular AF. Their predictable pharmacodynamic profiles allow fixed dose regimens with no need for routine monitoring. Hence, they make life much easier for patients and physicians. It is worth mentioning of some differences and controversial issues regarding those novel anticoagulant drugs (dabigatran, rivaroxaban, apixaban). Three large-scale phase III trials (RE-LY, ROCKET AF, and ARISTOTLE) compared these new anticoagulant agents with warfarin for prevention of stroke and systemic embolism (15, 25, 28). There are certain methodological and drug specific differences among the trials. A double -blind and double- dummy design were used in ROCKET AF and ARISTOTLE trials whereas RE-LY was designed as a randomized trial only to different dabigatran doses in a blinded fashion or to warfarin in an open label fashion. All these trials ended up with noninferiority of the study drugs compared with warfarin for prevention of stroke and systemic embolism. They also demonstrated favorable bleeding profiles compared with warfarin. Apixaban significantly decreased mortality in ARISTOTLE trial. However, anticoagulation levels achieved with warfarin were suboptimal in these trials. Mean percentage of study population with international normalized ratio (INR) in the therapeutic range was 64% in RELY, 55% in ROCKET AF and 62.2% in ARISTOTLE trials (15, 25, 28). This might have resulted in an overestimation of the beneficial effects of the study drugs, such as dabigatran, apixaban and rivaroxaban. To decide whether one drug is superior over the other is not possible at this stage because there is a lack of data comparing three drugs directly head to head in one randomized trial. Hence, each drug has its own pros and cons as well as side effect profiles. Even though it has a short half-life, once-daily regimen of rivaroxaban may be advantageous for patient compliance. This, however, also has raised the question whether full anticoagulation coverage and therapeutic INR values throughout the whole day may not be necessarily considered as a prerequisite for stroke prevention. Although not frequent, new anticoagulant agents may also interact with other drugs by interfering through the same metabolic pathways (17). For example, dabigatran interferes with amiodarone, verapamil and quinidine, which belong to the group of P-glycoprotein inhibitors. Therefore, dose reductions of dabigatran may be necessary in patients who take those certain drugs. Plasma level of rivaroxaban increases with inhibitors of CYP3A4 or P-glycoprotein (e.g. ketoconazole, erythromycine, clarithromycine, rintonavir), whereas plasma level decreases with inducers of CYP3A4 (e.g. rifampicin) (17). There are still limited data regarding the effects

of these drugs in populations who have not been adequately represented in the above mentioned trials (e.g. elderly patients, patients with renal insufficiency, or patients with liver impairment) (17). These agents are generally recommended to be cautiously used in patients with mild -to -moderate renal impairment and contraindicated in patients with severe renal insufficiency. There may be wide variations regarding the pharmacological and metabolic effects of new anticoagulants and there is currently no specific test for monitoring these drugs. Although some special coagulation tests may be applied to estimate the extent of thrombin or factor Xa inhibition, lack of standardized monitoring tests is still an important limitation for therapy with new anticoagulants especially in acute situations where measuring the anticoagulant effect is desirable (30). Monitoring the level of anticoagulation may also be necessary when there is a need for additional antithrombotic therapy such as after percutaneous coronary intervention where dual antiplatelet therapy should be additionally used. Another problematic issue related to new anticoagulant agents is the lack of any specific antidote to reverse the action of these drugs, especially in case of severe bleeding. Although not firmly established and supported by clinical data, activated prothrombin complex concentrate containing coagulation factors II, VII, IX, and X is suggested to reverse the effects of thrombin or factor Xa inhibitors (23). Recombinant factor VIIa is also another option to reverse the effects of factor Xa inhibitors, however its high cost and unproved efficacy to reduce bleeding are main drawbacks related with this agent (23). Research is ongoing for developing standardized tests and novel antidotes for the new anticoagulant agents.

Cost effectiveness of new anticoagulants is also an important issue. The cost of dabigatran and rivaroxaban therapy of one-month duration is approximately 25 times higher than that of warfarin in Turkey. Although the price of warfarin is relatively inexpensive, the costs of laboratory monitoring and potential complications are worth of mentioning. As such, clinical application, risks and benefits of new anticoagulants should be individually balanced. Cost effectiveness of new agents will be dependent on cost of individual drug, ability to achieve desired levels of anticoagulation with warfarin, individual patient risk for clotting and bleeding (more cost effective for higher risk patients) and performance of these drugs over warfarin.

It should also borne in mind that long-term follow up data of new anticoagulants are not yet available. Results of long-term follow up trials are needed to clarify the long term safety and efficacy of these drugs.

Conclusion

There is ongoing research with the aim of developing ideal anticoagulants for the prevention of thromboembolic events in patients with atrial fibrillation. Large-scale phase III trials are completed for dabigatran, which is an oral direct thrombin inhibitor and for rivaroxaban and apixaban which are oral factor Xa inhibitors. All these trials ended up with non-inferiority of these new anticoagulant drugs compared with warfarin for prevention of stroke and systemic embolism. They also showed a favorable bleeding profile compared with warfarin. Dabigatran and rivaroxaban have been approved by FDA for the prevention of stroke and systemic embolization in patients with non-valvular AF. However, there are still some controversies regarding application of these drugs in daily clinical practice. It seems that further research is necessary to resolve these controversies and clarify the clinical indications for each of these novel anticoagulants.

Conflict of interest: None declared.

Peer-review: Internally peer-reviewed.

Authorship contributions: Concept - O.E.; Design - O.E.; Supervision - O.E.; Resource - O.E., A.K.; Analysis &/or interpretation - O.E.; Literature search - O.E., A.K.; Writing - O.E, A.K.; Critical review - O.E.

References

- Wolf PA, Abbott RD, Kannel WB. Atrial fibrillation as an independent risk factor for stroke: the Framingham study. Stroke 1991; 22: 983-8. [CrossRef]
- Heeringa J, van der Kuip DA, Hofman A, Kors JA, van Herpen G, Stricker BH, et al. Prevalence, incidence and lifetime risk of atrial fibrillation: the Rotterdam study. Eur Heart J 2006; 27: 949-53.
 [CrossRef]
- Wann LS, Curtis AB, January CT, Ellenbogen KA, Lowe JE, Estes NA 3rd, et al. ACCF/AHA/HRS focused update on the management of patients with atrial fibrillation (Updating The 2006 Guideline): a report of the American College of Cardiology Foundation/American Heart Association Task Force on Practice Guidelines. Circulation 2011; 123: 104-23. [CrossRef]
- 4. Hart RG, Benavente O, McBride R, Pearce LA. Antithrombotic therapy to prevent stroke in patients with atrial fibrillation: a metaanalysis. Ann Intern Med 1999; 131: 492–501.
- Ansell J, Hirsh J, Hylek E, Jacobson A, Crowther M, Palareti G, et al. Pharmacology and management of the vitamin K antagonists: American College of Chest Physicians Evidence-Based Clinical Practice Guidelines (8th Edition). Chest 2008; 133: 160S-98S.
- Hylek EM, Evans-Molina C, Shea C, Henault LE, Regan S. Major hemorrhage and tolerability of warfarin in the first year of therapy among elderly patients with atrial fibrillation. Circulation 2007; 115: 2689–96. [CrossRef]
- Flaherty ML, Kissela B, Woo D, Kleindorfer D, Alwell K, Sekar P, et al. The increasing incidence of anticoagulant-associated intracerebral hemorrhage. Neurology 2007; 68: 116-21. [CrossRef]
- 8. Davis EM, Packard KA, Knezevich JT, Campbell JA. New and emerging anticoagulant therapy for atrial fibrillation and acute coronary syndrome. Pharmacotherapy 2011; 31: 975-1016. [CrossRef]
- Bassand JP. Review of atrial fibrillation outcome trials of oral anticoagulant and antiplatelet agents. Europace 2012; 14: 312-24. [CrossRef]
- Tran A, Cheng-Lai A. Dabigatran etexilate: the first oral anticoagulant available in the United States since warfarin. Cardiol Rev 2011; 19: 154-61. [CrossRef]

- 11. Stangier J, Clemens A. Pharmacology, pharmacokinetics, and pharmacodynamics of dabigatran etexilate, an oral direct thrombin inhibitor. Clin Appl Thromb Hemost 2009; 15: 9S-16S. [CrossRef]
- Viles-Gonzales JF, Fuster V, Halperin JL. New anticoagulants for prevention of stroke in patients with atrial fibrillation. J Cardiovasc Electrophysiol 2011; 22: 948-55. [CrossRef]
- Riva N, Lip GY. A new era for anticoagulation in atrial fibrillation. Which anticoagulant should we choose for long term prevention of thromboembolic complications in patients with atrial fibrillation? Pol Arch Med Wewn 2012; 122: 45-53.
- Hankey GJ, Eikelboom JW. Dabigatran etexilate: a new oral thrombin inhibitor. Circulation 2011; 123: 1436-50. [CrossRef]
- Connolly SJ, Ezekowitz MD, Yusuf S, Eikelboom J, Oldgren J, Parekh A, et al. Dabigatran versus warfarin in patients with atrial fibrillation. N Engl J Med 2009; 361: 1139-51. [CrossRef]
- Hohnloser SH, Oldgren J, Yang S, Wallentin L, Ezekowitz M, Reilly P, et al. Myocardial ischemic events in patients with atrial fibrillation treated with dabigatran or warfarin in the RE-LY (Randomized Evaluation of Long-Term Anticoagulation Therapy) trial. Circulation 2012; 125: 669-76. [CrossRef]
- Ahrens I, Lip GY, Peter K. New oral anticoagulant drugs in cardiovascular disease. Thromb Haemost 2010; 104: 49-60. [CrossRef]
- Diener HC, Connolly SJ, Ezekowitz MD, Wallentin L, Reilly PA, Yang S, et al. Dabigatran compared with warfarin in patients with atrial fibrillation and previous transient ischaemic attack or stroke: a subgroup analysis of the RE-LY trial. Lancet Neurol 2010; 9: 1157-63.
 [CrossRef]
- Nagarakanti R, Ezekowitz MD, Oldgren J, Yang S, Chernick M, Aikens TH, et al. Dabigatran versus warfarin in patients with atrial fibrillation – an analysis of patients undergoing cardioversion. Circulation 2011; 123: 131-6. [CrossRef]
- Oldgren J, Alings M, Darius H, Diener HC, Eikelboom J, Ezekowitz MD, et al. Risks for stroke, bleeding, and death in patients with atrial fibrillation receiving dabigatran or warfarin in relation to the CHADS2 score: a subgroup analysis of the RE-LY trial. Ann Intern Med 2011; 155: 660-7.

- Eikelboom JW, Wallentin L, Connolly SJ, Ezekowitz M, Healey JS, Oldgren J, et al. Risk of bleeding with 2 doses of dabigatran compared with warfarin in older and younger patients with atrial fibrillation: an analysis of the randomized evaluation of long-term anticoagulant therapy (RE-LY) trial. Circulation 2011; 123: 2363-72. [CrossRef]
- Roskell NS, Lip GY, Noack H, Clemens A, Plumb JM. Treatments for stroke prevention in atrial fibrillation: a network meta-analysis and indirect comparisons versus dabigatran etexilate. Thromb Haemost 2010; 104: 1106-15. [CrossRef]
- 23. Miesbach W, Seifried E. New direct oral anticoagulants current therapeutic options and treatment recommendations for bleeding complications. Thromb Haemost 2012; 108: 625-32. [CrossRef]
- 24. Perzborn E, Roehrig S, Straub A, Kubitza D, Mueck W, Laux V. Rivaroxaban: a new oral factor Xa inhibitor. Arterioscler Thromb Vasc Biol 2010; 30: 376-81. [CrossRef]
- 25. Patel MR, Mahaffey KW, Garg J, Pan G, Singer DE, Hacke W, et al. Rivaroxaban versus warfarin in nonvalvular atrial fibrillation. N Engl J Med 2011; 365: 883-91. [CrossRef]
- Schirmer SH, Baumhäkel M, Neuberger HR, Hohnloser SH, van Gelder IC, Lip GY, et al. Novel anticoagulants for stroke prevention in atrial fibrillation: current clinical evidence and future developments. J Am Coll Cardiol 2010; 56: 2067-76. [CrossRef]
- Raghavan N, Frost CE, Yu Z, He K, Zhang H, Humphreys WG, et al. Apixaban metabolism and pharmacokinetics after oral administration to humans. Drug Metab Dispos 2009; 37: 74–81. [CrossRef]
- Connolly SJ, Eikelboom J, Joyner C, Diener HC, Hart R, Golitsyn S, et al. Apixaban in patients with atrial fibrillation. N Engl J Med 2011; 364: 806-17. [CrossRef]
- 29. Granger CB, Alexander JH, McMurray JJ, Lopes RD, Hylek EM, Hanna M, et al. Apixaban versus warfarin in patients with atrial fibrillation. N Engl J Med 2011; 365: 981-92. [CrossRef]
- 30. Harenberg J, Marx S, Erdle S, Krämer R. Determination of the anticoagulant effects of new oral anticoagulants: an unmet need. Expert Rev Hematol 2012; 5: 107-13. [CrossRef]